Author's response to reviews

Title: Evaluation of Computerized Health Management Information System for Primary Health Care in rural India

Authors:

ANAND KRISHNAN (kanandiyer@yahoo.com)
BARIDALYNE NONGKYNRIH (baridalyne@gmail.com)
KAPIL YADAV (dr_kapilyadav@yahoo.co.in)
SATYAVIR SINGH (satyavir@medialabasia.in)
VIVEK GUPTA (drvivek.gupta@gmail.com)

Version: 6 Date: 12 October 2010

Author's response to reviews: see over
To
The Scientific Editor
BMC series journals

Subject: MS: 1250415418367419- Submission of revised manuscript

Dear Tim Shipley

Thank you for giving us the opportunity to revise the manuscript “Evaluation of Computerized Health Management Information System for Primary Health Care in rural India”. We are hereby submitting the revised version (11th October 2010). We have made the changes in tracked mode. We would be grateful if you will kindly consider it for publication in BMC Health Services Research.

With regards

Dr Baridalyne Nongkynrih
Associate Professor
Centre for Community Medicine
All India Institute of Medical Sciences
New Delhi -110029
Email: baridalyne@gmail.com